Author's response to reviews

Title: A Novel Oncolytic Adenovirus Targeting Cyclin E Overexpression Repressed Lung Tumor Growth

Authors:

Pei-Hsin Cheng (paisin.paisin@gmail.com)
Xiao-Mei Rao (x0rao001@exchange.louisville.edu)
Stephen L Wechman (swech01@exchange.louisville.edu)
Xiao-Feng Li (x0li0027@exchange.louisville.edu)
Kelly M. McMasters (kmmcma01@exchange.louisville.edu)
Heshan Sam Zhou (heshan.zhou@louisville.edu)

Version: 2 Date: 8 April 2015

Author's response to reviews:

1). Line numbering has been provided in the manuscript.
2). All the email addresses of the authors have been listed in the title page.
3). Consent section has been added in the end of the conclusion. Since this manuscript is a research article the consent is not applicable.